Author's response to reviews

Title: The addition of fluoxetine to cognitive behavioural therapy for youth depression (YoDA-C): study protocol for a randomised control trial

Authors:

Christopher G Davey (c.davey@unimelb.edu.au)
Andrew M Chanen (achanen@unimelb.edu.au)
Sue M Cotton (smcotton@unimelb.edu.au)
Sarah E Hetrick (shetrick@unimelb.edu.au)
Melissa J Kerr (Melissa.Kerr@mh.org.au)
Michael Berk (MIKEBE@BarwonHealth.org.au)
Olivia M Dean (olivia.dean@barwonhealth.org.au)
Kally Yuen (kally.yuen@unimelb.edu.au)
Mark Phelan (Mark.Phelan@mh.org.au)
Aswin Ratheesh (aswinr@unimelb.edu.au)
Miriam R Schäfer (Miriam.Schaefer@mh.org.au)
G Paul Amminger (amminger@unimelb.edu.au)
Alexandra G Parker (parkera@unimelb.edu.au)
Danijela Piskulic (daniela.piskulic@unimelb.edu.au)
Susy Harrigan (susy@unimelb.edu.au)
Andrew J Mackinnon (Andrew.Mackinnon@unimelb.edu.au)
Ben J Harrison (habj@unimelb.edu.au)
Patrick D McGorry (pmcgorry@unimelb.edu.au)

Version: 2
Date: 2 October 2014

Author's response to reviews: see over
2 October 2014

Dear Editor

We are pleased to provide a revised version of our manuscript. We have made the recommended formatting changes, comprising:

1. inclusion of the the email addresses of all authors on the title page;
2. removal of the word counts from the title page;
3. renaming the 'Design' section of the Abstract, 'Methods/Design';
4. renaming the 'Study Design' section 'Methods/Design';
5. inclusion of a list of abbreviations used and their meanings;
6. a statement in the Authors’ Contributions section that all authors read and approved the final version of the manuscript.

Yours sincerely,

Christopher Davey MBBS(Hons) PhD, FRANZCP
[on behalf of my co-authors]

Orygen Youth Health Research Centre & Centre for Youth Mental Health
The University of Melbourne
Parkville, Victoria 3052, Australia
e-mail: c.davey@unimelb.edu.au